Author's response to reviews

Title: Risk factors of diarrheal disease in under-five children among health extension model and non-model families in Sheko district rural community, Southwest Ethiopia

Authors:

Teklemichael Gebru (teklemichaelgebru@gmail.com)
Mohammed Taha (tahamohammed2009@gmail.com)
Wondwosen Kassahun (wondkyt@yahoo.com)

Version: 2 Date: 5 May 2013

Author's response to reviews:

May 04, 2013
To: BMC Public Health
Dear Editor:

We appreciate the opportunity to resubmit our research article “Risk factors of diarrhoeal disease in under-five children among health extension model and non-model families in Sheko district rural community, Southwest Ethiopia”. For the comments: 1) We made some changes to the formatting of the manuscript, and 2) We assured that the study protocol and the study was ethically approved by the health research and post graduate coordinating office, college of Public health and medical science of Jimma University. All comments by reviewers have been addressed, with corresponding changes made directly to the manuscript where appropriate. Accompanying this letter, please find a revised version of our manuscript.

Yours sincerely,
Teklemichael Gebru